Author’s response to reviews

Title: CD1d- and PJA2-related immune microenvironment differs between invasive breast carcinomas with and without a micropapillary feature.

Authors:

Naoki Kanomata (kanomata_7@med.kawasaki-m.ac.jp)

Junichi Kurebayashi (kure@med.kawasaki-m.ac.jp)

Yoshikazu Koike (ykoike@med.kawasaki-m.ac.jp)

Rin Yamaguchi (rin@med.kurume-u.ac.jp)

Takuya Moriya (tmoriya@med.kawasaki-m.ac.jp)

Version: 2 Date: 12 Dec 2018

Author’s response to reviews:

Responses to the comments:

Thank you very much for your careful review of our manuscript and helpful comments. Following the editorial comments, we have modified our manuscript.

-- Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

The email addresses for all authors were added to the title page.

-- Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

We add “Conclusions” section. The last two sentences in “Discussion” have been moved to “Conclusions”.

-- Please insert the heading 'Declarations' for the Declarations section, above the 'Abbreviations' section of your manuscript.
'Declarations' for the Declarations section was inserted above the 'Abbreviations' section.

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The role of funding was inserted.

-- Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

We corrected TM’s contribution.